Author's response to reviews

Title: A proposed method to investigate reliability throughout a questionnaire

Authors:

Tore Wentzel-Larsen (tore.wentzel-larsen@r-bup.no)
Tone M Norekvål (tone.norekval@helse-bergen.no)
Bjørg Ulvik (bjorg.ulvik@hib.no)
Ottar Nygård (ottar.nygard@helse-bergen.no)
Are H Pripp (are.hugo.pripp@oslo-universitetssykehus.no)

Version: 3 Date: 2 September 2011

Author's response to reviews:

Dear Editor Adrian Aldcroft;

Thank you for your interest in our research and help with the revision and review of the manuscript. One of the reviewers asked us to mention the literature on item response theory. We have added a new paragraph in the discussion that relate this interesting field to our research. The manuscript has also been checked regarding format and spelling errors. Especially, the format of the references has been updated. Please find a revised version of the manuscript and a document with "track changes". The format changes to the references are not shown with "track changes" due to technical reasons.

We look forward to your assessment of the revised version. The suggestions and comments by the reviewers were very helpful.

Sincerely;

Are Hugo Pripp